Author's response to reviews

Title: Resolution of biliary stricture after living donor liver transplantation in a child by PTCD: a case report

Authors:

Gabriel Putzer (gabriel.putzer@i-med.ac.at)
Peter Paal (peter.paal@i-med.ac.at)
Andreas P. Chemelli (andreas.chemelli@i-med.ac.at)
Walter Mark (walter.mark@i-med.ac.at)
Wolfgang Lederer (wolfgang.lederer@i-med.ac.at)
Franz J. Wiedermann (franz.wiedermann@i-med.ac.at)

Version: 2 Date: 4 April 2013

Author's response to reviews:

Dear Editor,

please find attached the revised manuscript entitled "Resolution of biliary stricture after living donor liver transplantation in a child by PTCD: a case report" by Putzer G, Paal P, Chemelli AP, Mark W, Lederer W, Wiedermann FJ.

The manuscript has been revised according the recommendations of the reviewers.

All important changes of the manuscript are indicated with underlines.

We included a description of the former surgical history, available histopathology, and a short description of the PTCD procedure.

The manuscript was corrected by a native English speaking lecturer.

We look forward to hearing from you.

Best regards.

Yours sincerely,

Franz J. Wiedermann, MD, DEAA, FCCP